Author's response to reviews

Title: The incremental treatment of ESRD: A low-protein diet combined with weekly hemodialysis may be beneficial for selected patients.

Authors:

Stefania Caria SC (cariastefan@tiscali.it)
Adamasco Cupisti AP (adamasco.cupisti@med.unipi.it)
Giovanna Sau GS (giovannasau@aob.it)
Piergiorgio Bolasco PB (pg.bolasco@tin.it)

Version: 3 Date: 15 July 2014

Author's response to reviews:

Coming from the experience of Mitch and than from the studies of Locatelli et Al, in the time we have simplified the previous models in order to have the best advantages from hypoproteic diet and its resulting slowing loss of kidney residual function. This research could be the point of departure towards progressively tailored dialysis dose on the patient's metabolic needs while maintaining a good nutritional status.